Author's response to reviews

Title: Delayed treatment of Basilar Thrombosis in a patient with a Basilar Aneurysm: A Case Report

Authors:

Tamer Fakhouri (TFakhouri@student.uchc.edu)
Louise D McCullough (lmccullough@uchc.edu)

Version: 3 Date: 27 May 2008

Author's response to reviews:

It was noted that the publication still needed several minor modifications prior to review. These are stated here and have been corrected in the manuscript.

1. Please include co-author's affiliation in the title page-now listed

2. Please include all the abbreviations used in the manuscript and their meanings in the 'abbreviations list'. All abbreviations are listed. Strength testing is listed as 2/5 or 3/5 as a standard neurological score where 5/5 is maximal strength. Please let us know what (if anything) needs further details.